Author's response to reviews

Title: Tracking Family Medicine Graduates. Where do they go, what services do they provide and whom do they see?

Authors:

R. Liisa Jaakkimainen (liisa.jaakkimainen@ices.on.ca)
Susan E. Schultz (sue.schultz@ices.on.ca)
Richard H. Glazier (Rick.Glazier@ices.on.ca)
Caroline Abrahams (Caroline.Abrahams@utoronto.ca)
Sarita Verma (Sarita.Verma@utoronto.ca)

Version: 3 Date: 30 January 2012

Author's response to reviews: see over
Dear Editor of BMC Family Practice,

I am submitting a revision to a paper titled “Tracking Family Medicine Graduates. Where do they go, what services do they provide and whom do they see?” for consideration of publication in BMC Family Practice.

Thank you again for the comments from both reviewers. These comments were again helpful and we have revised the manuscript based on their suggestions. We also did some minor grammatical editing to this version of the paper.

Specifically we added more to the discussion on the usefulness of health administrative data for physician resource planning. We also included a statement about health administrative data in the conclusions. We also re-worded the abstract to mirror the text in the paper.

I appreciate your consideration of the revised version of this research article for publication. If you need any more information, please let me know.

Yours sincerely,

R. Liisa Jaakkimainen MD, MSc, CCFP (Corresponding Author)
Associate Professor, Family and Community Medicine, University of Toronto
Scientist, Institute for Clinical Evaluative Sciences
Family Physician, Sunnybrook Health Sciences Centre

Institute for Clinical Evaluative Sciences
2075 Bayview Ave, G wing
Toronto, Ontario
M4N 3M5
Phone: 416-480-4055 ext 3838
Email: liisa.jaakkimainen@ices.on.ca